%File content.tex, 9 September, 1992
%\magnification=\magstep1
\input ij.sty
\input mssymb
\def\ {\thinspace}
\lineskiplimit=100pt 
\lineskip=7pt %This controls the spacing between indexitems of
%both kinds
\tenpoint
\null\vskip 3cm

\def\index\indexvolume#1\indexyear#2\endindextitle{
\rm\parindent=0pt%\overfullrule=0pt
\centerline{ISRAEL JOURNAL OF MATHEMATICS}
\vskip0.4cm
\centerline{TABLE OF CONTENTS, VOLUME #1\unskip, #2\unskip}
\vskip1.2cm
\frenchspacing}
\def\endindex{}

\def\indexitem\indexauthor#1\indextitle#2\indexpage#3\endindexitem
{\halign {\normalbaselines%
\vbox{\hsize=11cm\hangindent=2em\hangafter=1 ##}%
&\hbox to 1.3cm{\hfill ##}\cr%
{\smc #1\unskip}\quad{\sl #2} & #3 \cr}\par}
\def\indexauthor{}
\def\endindexitem{}
\def\indexpage{}
\def\indextitle{}

\def\indexsee\indexauthor#1\partner#2\endindexitem
{{\smc #1\unskip\enspace}{\sl See }{\smc #2\unskip}\par}

\index
\indexvolume 105
\indexyear 1998
\endindextitle

\indexitem 
\indexauthor Abarbanel, J.
\indextitle On the Schur multiplier of augmented algebras
\indexpage 197
\endindexitem

\indexitem
\indexauthor Alekseevsky, D., A.~Kriegl, P.~W.~Michor and M.~Losik\break
\indextitle Choosing roots of polynomials smoothly
\indexpage 203
\endindexitem

\indexitem
\indexauthor Ara, P., K.~R.~Goodearl, K.~C.~O'Meara and E.~Pardo\break
\indextitle Separative cancellation for projective modules over
exchange rings
\indexpage 105
\endindexitem

\indexitem
\indexauthor Campi, S., A.~Colesanti and P.~Gronchi
\indextitle Blaschke-decomposable convex bodies
\indexpage 185
\endindexitem

\indexsee
\indexauthor Colesanti, A.
\partner S.~Campi
\endindexitem

\indexitem
\indexauthor Corvaja, P. and U.~Zannier
\indextitle Values of rational functions on non-hilbertian fields
and a question of Weissauer
\indexpage 323
\endindexitem

\indexitem
\indexauthor Deville, R., V.~Fonf and P.~H\'ajek
\indextitle Analytic and polyhedral\break approximation of convex bodies
in separable polyhedral Banach\break spaces
\indexpage 139
\endindexitem

\indexsee
\indexauthor Fonf, V.
\partner R.~Deville
\endindexitem

\indexitem
\indexauthor Friedgut, E. and J.~Kahn
\indextitle On the number of copies of one\break hypergraph in another
\indexpage  251
\endindexitem 

\indexsee
\indexauthor Goodearl, K.~R.
\partner P.~Ara
\endindexitem

\indexsee
\indexauthor Gronchi, P.
\partner S.~Campi
\endindexitem

\indexitem
\indexauthor Guenais, M.
\indextitle Une majoration de la multiplicit\'e spectrale
d'op\'erateurs associ\'es \`a des cocycles r\'eguliers
\indexpage 263
\endindexitem

\indexsee
\indexauthor H\'ajek, P.
\partner R.~Deville
\endindexitem

\indexsee  
\indexauthor Kahn, J.
\partner E.~Friedgut
\endindexitem

\indexitem
\indexauthor Kenyon, R.
\indextitle Tilings and discrete Dirichlet problems
\indexpage 61
\endindexitem

\indexitem
\indexauthor Kierstead, H.~A.
\indextitle On-line coloring $k$-colorable graphs
\indexpage 93
\endindexitem

\indexitem
\indexauthor Komorowski, R.~A. and N.~Tomczak-Jaegermann
\indextitle Erratum to: ``Banach spaces without local unconditional
structure''
\indexpage 85
\endindexitem

\indexsee
\indexauthor Kriegl, A.
\partner D.~Alekseevsky
\endindexitem

\indexitem
\indexauthor L\'evai, L.
\indextitle Linear core conditions in residually finite groups
\indexpage 337
\endindexitem

\indexitem
\indexauthor Levin, M.
\indextitle Certain finite dimensional maps and their application to
hyperspaces
\indexpage 257
\endindexitem

\indexsee
\indexauthor Losik, M.
\partner D.~Alekseevsky
\endindexitem

\indexitem 
\indexauthor Meiri, D.
\indextitle Entropy and uniform distribution of orbits in ${\Bbb
T}^d$
\indexpage 155
\endindexitem

\indexsee
\indexauthor Michor, P.~W.
\partner D.~Alekseevsky
\endindexitem

\indexitem
\indexauthor Nagy, P.~T. and K.~Strambach
\indextitle Loops, their cores and symmetric spaces
\indexpage 285
\endindexitem

\indexitem
\indexauthor Novo, S. and R.~Obaya 
\indextitle On the dynamical behaviour of an almost\break periodic linear
system with an ergodic 2-sheet studied by\break R.~Johnson
\indexpage 235
\endindexitem

\indexsee
\indexauthor Obaya, R.
\partner S.~Novo
\endindexitem

\indexsee
\indexauthor O'Meara, K.~C.
\partner P.~Ara
\endindexitem

\indexsee
\indexauthor Pardo, E.
\partner P.~Ara
\endindexitem

\indexitem
\indexauthor Peretz, R.
\indextitle The geometry of the asymptotics of polynomial maps
\indexpage 1
\endindexitem

\indexsee
\indexauthor Strambach, K.
\partner P.~T.~Nagy
\endindexitem

\indexsee
\indexauthor Tomczak-Jaegermann, N.
\partner R.~A.~Komorowski
\endindexitem

\indexsee
\indexauthor Zannier, U.
\partner P.~Corvaja
\endindexitem

\endindex
%End of file, content.tex
